Author's response to reviews

Title: Obesity, Physical Activity, and the Urban Environment: Public Health Research Needs

Authors:

Russell P. Lopez (rptlopez@bu.edu)
H. Patricia Hynes (hph@bu.edu)

Version: 4 Date: 11 September 2006

Author's response to reviews:

Dear Dr. Grandjean:

I have reformatted the article as you outlined and responded to your comments. The only revision you may want to note is that I added a brief definition of the term "gridded" which refers to small blocks of streets set at right angles.

Again, thank you for your time spent on our article.

Russ Lopez